Author's response to reviews

Title: Does a decision aid improve informed choice in mammography screening?
Study protocol for a randomized controlled trial

Authors:

Maren Reder (maren.reder@uni-bielefeld.de)
Petra Kolip (petra.kolip@uni-bielefeld.de)

Version: 2 Date: 8 August 2014

Author's response to reviews: see over
Dear Editorial Team:

The line numbering of the manuscript is now continuous. The authors’ affiliations have additionally been added behind the author names.

The funding application process that our proposal underwent with the Barmer GEK was as follows: We sent a proposal of the study to the Barmer GEK, which was evaluated by experts within the Barmer GEK and accepted by the Barmer GEK. This funding application process did not involve full, independent peer-review.

Sincerely,
Maren Reder